Author's response to reviews

Title: Effectiveness of Automated Locomotor Training in Patients with Acute Incomplete Spinal Cord Injury: A Randomized Controlled Multicenter Trial

Authors:

Markus Wirz (mwirz@paralab.balgrist.ch)  
Carolien Bastiaenen (chg.bastiaenen@maastrichtuniversity.nl)  
Rob de Bie (ra.debie@maastrichtuniversity.nl)  
Volker Dietz (vdietz@paralab.balgrist.ch)

Version: 3 Date: 24 May 2011

Author's response to reviews:

Dear Editor

We revised the submission according to your mail from May 19th. The changes made are listed here:

• Author list revised: It does now match the list in the submission system.
• Abstract: the abstract in the submission system and in the manuscript are now identical.
• Structure: we did not include a heading ‘Discussion’ since it is not applicable.
• Highlighted changes from revisions removed.
• Figure in low resolution removed from the manuscript.

We hope that the manuscript does now conform to the requirements. Thank you again for your support. If you have further questions, please do not hesitate to contact us.

Sincerely,
Markus Wirz, for the authors